\begin{table}%t4
\par
%\centering
\par
\caption{\label{tab:gf1}Weighted absorption oscillator strengths ($gf$).}
\begin{tabular}{lccccc}
 \hline\hline 
\noalign{\smallskip}
\multicolumn{2}{l}{Upper level} & \multicolumn{3}{c}{Lower level}   \\
 $u$        & $^3$P$_{2}$$^{\rm e}$   & $^3$P$_{1}^{\rm e}$   & $^3$P$_{0}^{\rm e}$   & $^1$D$_{2}^{\rm e}$   & $^1$S$_{0}^{\rm e}$  \\[-3.2mm] \\ 
\hline                                 
\par
34--$^3$P$_{2}^{\rm o}$   & 202.42   & 207.75   &  --  & 219.17   &  -- \\[-3.2mm] \\ 
\hline 
48CT   & 0.13 & 3.3 $\times$ 10$^{-2}$ &  --  & 5.5 $\times$ 10$^{-3}$ &  --  \\ 
48C   & 0.13 & 3.2 $\times$ 10$^{-2}$ &  --   & 5.2 $\times$ 10$^{-3}$ &  --   \\ 
13C   & 0.16 & 3.6 $\times$ 10$^{-2}$ &  --   & 7.6 $\times$ 10$^{-3}$ &  --   \\ 
DT98   & 7.4 $\times$ 10$^{-2}$ & 1.2 $\times$ 10$^{-2}$ &  --   & 5.8 $\times$ 10$^{-3}$ &  --   \\ 
AK03   & 0.10 & 2.9 $\times$ 10$^{-2}$ &  --   & 2.7 $\times$ 10$^{-3}$ &  --   \\ 
BDE02   & 0.11 & 3.1 $\times$ 10$^{-2}$ &  --   & 3.5 $\times$ 10$^{-3}$ &  --   \\ 
F86   & 0.11 & 2.9 $\times$ 10$^{-2}$ &  --   &  --   &  --   \\ 
\hline\\[-3.2mm]
35--$^3$D$_{3}^{\rm o}$   & 201.12   &  --  &  --  & 217.64   &  -- \\[-3.2mm] \\ 
\hline\\[-3.2mm]
48CT   & 2.2 $\times$ 10$^{-2}$ &  --  &  --  & 3.3 $\times$ 10$^{-3}$ &  --  \\ 
48C   & 2.2 $\times$ 10$^{-2}$ &  --   &  --   & 3.1 $\times$ 10$^{-3}$ &  --   \\ 
13C   & 2.1 $\times$ 10$^{-2}$ &  --   &  --   & 2.5 $\times$ 10$^{-3}$ &  --   \\ 
DT98   & 5.0 $\times$ 10$^{-2}$ &  --   &  --   & 8.8 $\times$ 10$^{-3}$ &  --   \\ 
AK03   & 1.9 $\times$ 10$^{-2}$ &  --   &  --   & 2.5 $\times$ 10$^{-3}$ &  --   \\ 
BDE02   & 3.2 $\times$ 10$^{-2}$ &  --   &  --   & 2.9 $\times$ 10$^{-3}$ &  --   \\ 
F86   & 5.0 $\times$ 10$^{-2}$ &  --   &  --   &  --   &  --   \\ 
\hline\\[-3.2mm]
36--$^1$F$_{3}^{\rm o}$   & 190.38   &  --  &  --  & 205.12   &  -- \\[-3.2mm] \\ 
\hline\\[-3.2mm]
48CT   & 3.2 $\times$ 10$^{-2}$ &  --  &  --  & 3.6 $\times$ 10$^{-4}$ &  --  \\ 
48C   & 2.7 $\times$ 10$^{-2}$ &  --   &  --   & 4.1 $\times$ 10$^{-4}$ &  --   \\ 
13C   & 3.3 $\times$ 10$^{-2}$ &  --   &  --   & 2.1 $\times$ 10$^{-3}$ &  --   \\ 
DT98   & 3.1 $\times$ 10$^{-4}$ &  --   &  --   & 2.2 $\times$ 10$^{-2}$ &  --   \\ 
AK03   & 2.2 $\times$ 10$^{-2}$ &  --   &  --   & 7.4 $\times$ 10$^{-4}$ &  --   \\ 
BDE02   & 1.7 $\times$ 10$^{-2}$ &  --   &  --   & 5.1 $\times$ 10$^{-4}$ &  --   \\ 
F86   & 1.4 $\times$ 10$^{-2}$ &  --   &  --   &  --   &  --   \\ 
\hline\\[-3.2mm]
37--$^3$P$_{1}^{\rm o}$   & 188.30   & 192.90   & 193.51   & 202.71   & 222.10  \\[-3.2mm] \\ 
\hline\\[-3.2mm]
48CT   & 1.5 & 2.5 $\times$ 10$^{-3}$ & 2.5 $\times$ 10$^{-2}$ & 0.18 & 2.1 $\times$ 10$^{-2}$ \\ 
48C   & 1.6 & 3.7 $\times$ 10$^{-2}$ & 2.7 $\times$ 10$^{-3}$ & 4.2 $\times$ 10$^{-2}$ & 1.0 $\times$ 10$^{-2}$ \\ 
13C   & 1.5 & 2.9 $\times$ 10$^{-3}$ & 6.5 $\times$ 10$^{-2}$ & 0.26 & 3.0 $\times$ 10$^{-2}$ \\ 
DT98   & 1.3 & 0.20 & 2.7 $\times$ 10$^{-2}$ & 7.9 $\times$ 10$^{-3}$ & 1.5 $\times$ 10$^{-2}$ \\ 
AK03   & 0.76 & 0.27 & 7.6 $\times$ 10$^{-2}$ & 2.6 $\times$ 10$^{-2}$ & 1.5 $\times$ 10$^{-3}$ \\ 
BDE02   & 1.1 & 0.38 & 0.11 & 2.5 $\times$ 10$^{-2}$ & 1.4 $\times$ 10$^{-3}$ \\ 
F86   & 1.6 &  --   & 2.1 $\times$ 10$^{-2}$ & 0.35 & 4.1 $\times$ 10$^{-2}$ \\ 
\hline\\[-3.2mm]
38--$^3$P$_{2}^{\rm o}$   & 188.22   & 192.81   &  --  & 202.61   &  -- \\[-3.2mm] \\ 
\hline\\[-3.2mm]
48CT   & 2.6 & 0.56 &  --  & 5.0 $\times$ 10$^{-2}$ &  --  \\ 
48C   & 2.6 & 0.58 &  --   & 4.9 $\times$ 10$^{-2}$ &  --   \\ 
13C   & 2.6 & 0.57 &  --   & 4.7 $\times$ 10$^{-2}$ &  --   \\ 
DT98   & 2.7 & 0.61 &  --   & 4.5 $\times$ 10$^{-2}$ &  --   \\ 
AK03   & 3.4 & 0.69 &  --   & 4.8 $\times$ 10$^{-2}$ &  --   \\ 
BDE02   & 4.0 & 0.79 &  --   & 4.4 $\times$ 10$^{-2}$ &  --   \\ 
F86   & 2.9 & 0.60 &  --   & 5.1 $\times$ 10$^{-2}$ &  --   \\ 
\hline\\[-3.2mm]
39--$^3$S$_{1}^{\rm o}$   & 187.45   & 192.01   & 192.62   & 201.72   & 220.92  \\[-3.2mm] \\
 \hline\\[-3.2mm]
48CT   & 4.1 $\times$ 10$^{-2}$ & 0.45 & 0.25 & 1.0 & 4.3 $\times$ 10$^{-2}$ \\ 
48C   & 7.7 $\times$ 10$^{-3}$ & 0.50 & 0.33 & 1.0 & 4.7 $\times$ 10$^{-2}$ \\ 
13C   & 0.13 & 0.47 & 0.26 & 0.90 & 4.2 $\times$ 10$^{-2}$ \\ 
DT98   & 0.32 & 0.48 & 0.43 & 8.0 $\times$ 10$^{-2}$ & 3.6 $\times$ 10$^{-2}$ \\ 
AK03   & 0.97 & 0.68 & 0.60 & 0.35 & 6.1 $\times$ 10$^{-4}$ \\ 
BDE02   & 1.1 & 0.83 & 0.72 & 0.39 & 7.0 $\times$ 10$^{-5}$ \\ 
F86   & 0.24 & 0.49 & 0.22 & 1.1 & 4.2 $\times$ 10$^{-2}$ \\ 
\hline 
40--$^3$P$_{0}^{\rm o}$   &  --  & 189.00   &  --  &  --  &  --  \\ 
\hline
48CT   &  --  & 0.69 &  --  &  --  &  --  \\ 
48C   &  --   & 0.70 &  --   &  --   &  --   \\ 
13C   &  --   & 0.70 &  --   &  --   &  --   \\ 
DT98   &  --   & 0.71 &  --   &  --   &  --   \\ 
AK03   &  --   & 0.87 &  --   &  --   &  --   \\ 
BDE02   &  --   & 1.0 &  --   &  --   &  --   \\ 
F86   &  --   & 0.75 &  --   &  --   &  --   \\ 
\hline\\[-3.2mm]
41--$^3$P$_{1}^{\rm o}$   & 184.70   & 189.12   & 189.71   & 198.54   & 217.11 \\[-3.2mm]   \\ 
\hline\\[-3.2mm]
48CT   & 0.12 & 0.59 & 0.52 & 0.69 & 7.7 $\times$ 10$^{-2}$ \\ 
48C   & 0.12 & 0.52 & 0.47 & 0.85 & 8.6 $\times$ 10$^{-2}$ \\ 
13C   & 8.6 $\times$ 10$^{-2}$ & 0.59 & 0.49 & 0.77 & 9.9 $\times$ 10$^{-2}$ \\ 
DT98   & 0.12 & 0.36 & 0.35 & 0.11 & 0.10 \\ 
AK03   & 2.8 $\times$ 10$^{-2}$ & 9.8 $\times$ 10$^{-2}$ & 0.24 & 2.2 & 9.5 $\times$ 10$^{-2}$ \\ 
BDE02   & 3.7 $\times$ 10$^{-2}$ & 0.12 & 0.26 & 2.5 & 8.6 $\times$ 10$^{-2}$ \\ 
F86   & 0.16 & 0.76 & 0.66 &  --   & 6.1 $\times$ 10$^{-2}$ \\ 
\hline\\[-3.2mm]
42--$^3$D$_{3}^{\rm o}$   & 180.40   &  --  &  --  & 193.58   &  -- \\[-3.2mm]  \\ 
\hline\\[-3.2mm]
48CT   & 4.6 &  --  &  --  & 2.8 $\times$ 10$^{-3}$ &  --  \\ 
48C   & 4.7 &  --   &  --   & 3.4 $\times$ 10$^{-3}$ &  --   \\ 
13C   & 4.8 &  --   &  --   & 2.6 $\times$ 10$^{-3}$ &  --   \\ 
DT98   & 4.7 &  --   &  --   & 2.8 $\times$ 10$^{-4}$ &  --   \\ 
AK03   & 5.3 &  --   &  --   & 3.9 $\times$ 10$^{-3}$ &  --   \\ 
BDE02   & 6.2 &  --   &  --   & 5.2 $\times$ 10$^{-3}$ &  --   \\ 
F86   & 5.3 &  --   &  --   &  --   &  --   \\ 
\hline\\[-3.2mm]
43--$^3$D$_{2}^{\rm o}$   & 178.06   & 182.16   &  --  & 190.88   &  -- \\[-3.2mm]  \\ 
\hline\\[-3.2mm]
48CT   & 0.65 & 2.6 &  --  & 8.3 $\times$ 10$^{-2}$ &  --  \\ 
48C   & 0.67 & 2.6 &  --   & 8.1 $\times$ 10$^{-2}$ &  --   \\ 
13C   & 0.68 & 2.6 &  --   & 7.8 $\times$ 10$^{-2}$ &  --   \\ 
DT98   & 0.67 & 2.5 &  --   & 0.12 &  --   \\ 
AK03   & 0.62 & 2.9 &  --   & 0.15 &  --   \\ 
BDE02   & 0.72 & 3.4 &  --   & 0.18 &  --   \\ 
F86   & 0.75 & 2.9 &  --   & 0.13 &  --   \\ 
\hline\\[-3.2mm]
44--$^3$D$_{1}^{\rm o}$   & 176.55   & 180.59   & 181.13   & 189.16   & 205.95 \\[-3.2mm]   \\ 
\hline\\[-3.2mm]
48CT   & 3.4 $\times$ 10$^{-2}$ & 0.81 & 1.1 & 6.0 $\times$ 10$^{-3}$ & 2.0 $\times$ 10$^{-4}$ \\ 
48C   & 3.6 $\times$ 10$^{-2}$ & 0.82 & 1.2 & 6.1 $\times$ 10$^{-3}$ & 2.0 $\times$ 10$^{-4}$ \\ 
13C   & 3.7 $\times$ 10$^{-2}$ & 0.83 & 1.2 & 4.0 $\times$ 10$^{-3}$ & 2.4 $\times$ 10$^{-4}$ \\ 
DT98   & 3.8 $\times$ 10$^{-2}$ & 0.81 & 1.2 & 6.0 $\times$ 10$^{-3}$ & 3.5 $\times$ 10$^{-4}$ \\ 
AK03   & 3.3 $\times$ 10$^{-2}$ & 0.91 & 1.3 & 6.9 $\times$ 10$^{-2}$ & 1.1 $\times$ 10$^{-2}$ \\ 
BDE02   & 3.9 $\times$ 10$^{-2}$ & 1.1 & 1.5 & 5.5 $\times$ 10$^{-2}$ & 3.2 $\times$ 10$^{-3}$ \\ 
F86   & 4.0 $\times$ 10$^{-2}$ & 0.92 & 1.3 &  --   &  --   \\ 
\hline\\[-3.2mm]
45--$^1$D$_{2}^{\rm o}$   & 172.74   & 176.61   &  --  & 184.79   &  -- \\[-3.2mm]  \\ 
\hline\\[-3.2mm]
48CT   & 4.8 $\times$ 10$^{-4}$ & 0.15 &  --  & 2.6 &  --  \\ 
48C   & 5.0 $\times$ 10$^{-4}$ & 0.15 &  --   & 2.7 &  --   \\ 
13C   & 5.0 $\times$ 10$^{-4}$ & 0.15 &  --   & 2.7 &  --   \\ 
DT98   & 6.2 $\times$ 10$^{-4}$ & 0.19 &  --   & 2.6 &  --   \\ 
AK03   & 1.5 $\times$ 10$^{-2}$ & 0.29 &  --   & 2.3 &  --   \\ 
BDE02   & 1.3 $\times$ 10$^{-2}$ & 0.29 &  --   & 3.1 &  --   \\ 
F86   &  --   & 0.20 &  --   & 3.3 &  --   \\ 
\hline\\[-3.2mm]
46--$^1$F$_{3}^{\rm o}$   & 168.34   &  --  &  --  & 179.76   &  -- \\[-3.2mm]  \\
 \hline\\[-3.2mm]
48CT   & 8.2 $\times$ 10$^{-3}$ &  --  &  --  & 5.0 &  --  \\ 
48C   & 9.2 $\times$ 10$^{-3}$ &  --   &  --   & 5.1 &  --   \\ 
13C   & 7.5 $\times$ 10$^{-3}$ &  --   &  --   & 5.2 &  --   \\ 
DT98   & 3.8 $\times$ 10$^{-3}$ &  --   &  --   & 5.1 &  --   \\ 
AK03   & 1.9 $\times$ 10$^{-2}$ &  --   &  --   & 6.3 &  --   \\ 
BDE02   & 1.7 $\times$ 10$^{-2}$ &  --   &  --   & 7.1 &  --   \\ 
F86   &  --   &  --   &  --   & 5.6 &  --   \\

\end{tabular}
\tablefoot{Only transitions
between the five levels of the ground configuration and the 3s$^2$3p$^3$3d
levels are shown. For each upper level, we list the index and $LSJ$ from 
 Table~\ref{tab:energies1}, together with the wavelength of the 
transition in \AA.
We then list the $gf$ vales from our benchmark calculation with 
TEC (48CT) and without TEC (48C), those from our target basis (13C),
together with a selection of values from the literature.  
DT98: \cite{deb_tayal:98};
AK03: \cite{aggarwal_keenan:03b,aggarwal_keenan:03a};
BDE02: \cite{bhatia_etal:02};
F86: \cite{fawcett:86}.}
\end{table}